Author's response to reviews

Title: Femoral nerve block-sciatic nerve block vs. femoral nerve block-local infiltration analgesia for total knee arthroplasty: a randomized controlled trial

Authors:

Mari Nagafuchi (marimatching2@yahoo.co.jp)
Tomoyuki Sato (tomoyukisatoh@seagreen.ocn.ne.jp)
Takahiro Sakuma (taka1hiro1@gmail.com)
Akemi Uematsu (uematsu2@bk2.so-net.ne.jp)
Hiromasa Hayashi (edelweiss101gou@hotmail.co.jp)
Hidenori Tanikawa (adriatic123sea@gmail.com)
Kazunari Okuma (ookuma.kazu@gmail.com)
Akira Hashiuchi (hashiuchi@m2.pbc.ne.jp)
Junya Oshida (oswillkommen@ybb.ne.jp)
Hiroshi Morisaki (morisaki@z8.keio.jp)

Version: 2
Date: 7 October 2014

Author's response to reviews: see over
Dear Dr. Gaiser:

Thank you for giving us the opportunity to correct the formatting of our manuscript as per the author guidelines. We have modified the manuscript accordingly and have detailed our actions and responses below:

*Line numbering:*

*Please revise your manuscript to include line and page numbers. Authors are asked to ensure that line numbering is included in the main text file of their manuscript at the time of submission to facilitate peer-review.*

We have inserted continuous line numbers in the manuscript and removed page breaks as requested in the author guidelines.

The manuscript has now been resubmitted to your journal. We look forward to a positive response.

Sincerely,

Tomoyuki Sato
Department of Anesthesiology, Saitama City Hospital
Mimuro 2460, Saitama Midori-ku, Saitama, 336-8522 Japan
Tel: +81-48-873-4111
Fax: +81-48-873-5451
E-mail: tomoyukisatoh@seagreen.ocn.ne.jp